ORIGINAL RESEARCH



Psychomotor Vigilance Performance in Participants with Excessive Daytime Sleepiness in Obstructive Sleep Apnea or Narcolepsy Compared with SAFTE-FAST Model Predictions

Jaime K. Devine (b) · Lindsay Schwartz (b) · Steven Hursh (b) · Jerryll Asin (b) · Nicolaas de Vries · Patty E. Vonk (b) · Annemiek Vermeeren (b) · Claire E. H. M. Donjacour (b) · Frederick Vinckenbosch (b) · Johannes G. Ramaekers (b) · Hennie Janssen (b) · Grace Wang · Dan Chen · Lawrence P. Carter · Sebastiaan Overeem (b) · Gert Jan Lammers (b)

Received: July 29, 2022 / Accepted: November 9, 2022 / Published online: December 10, 2022 \odot The Author(s) 2022

ABSTRACT

Introduction: Excessive daytime sleepiness (EDS) associated with narcolepsy or obstructive sleep apnea (OSA) can impair vigilance/attention. Solriamfetol, a dopamine/norepinephrine reuptake inhibitor, is approved to treat EDS associated with narcolepsy (75–150 mg/day) or OSA (37.5–150 mg/day). The analysis reported here explored the use of the Sleep, Activity,

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s40120-022-00425-w.

J. K. Devine (⊠) · L. Schwartz · S. Hursh Institutes for Behavior Resources, Inc., 2104 Maryland Avenue, Baltimore, MD 21218, USA e-mail: jdevine@ibrinc.org

S. Hursh Johns Hopkins University School of Medicine, Baltimore, MD, USA

J. Asin Center for Sleep Medicine Amphia, Breda, Netherlands

N. de Vries

Department of Otorhinolaryngology, Head and Neck Surgery, Faculty of Medicine and Health Sciences, Antwerp University Hospital, Antwerp, Belgium

N. de Vries

Department of Oral Kinesiology, Academic Centre for Dentistry Amsterdam, Amsterdam, Netherlands Fatigue, and Task Effectiveness (SAFTE) model (used in transport industries to model performance based on accumulated sleep and circadian variability) as a substitute for healthy controls using psychomotor vigilance task (PVT) data collected during clinical studies. Methods: Data were analyzed from two phase 2 studies of solriamfetol in adults with OSA (NCT02806895, EudraCT 2015-003930-28) or narcolepsv (NCT02806908, EudraCT 2015-003931-36). Participants were randomly assigned 1:1 to solriamfetol 150 mg/day (3 days) followed by 300 mg/day (4 days), or placebo (7 days), then crossed over to the other treatment.

N. de Vries MOVE Research Institute Amsterdam, University of Amsterdam and VU University Amsterdam, Amsterdam, Netherlands

N. de Vries · P. E. Vonk Onze Lieve Vrouwe Gasthuis (OLVG), Amsterdam, Netherlands

A. Vermeeren · F. Vinckenbosch · J. G. Ramaekers Faculty of Psychology and Neuroscience, Maastricht University, Maastricht, Netherlands

C. E. H. M. Donjacour · G. J. Lammers Sleep–Wake Centre SEIN, Heemstede, Netherlands

H. Janssen · S. Overeem Sleep Medicine Center Kempenhaeghe, Heeze, Netherlands

G. Wang · D. Chen Jazz Pharmaceuticals, Palo Alto, CA, USA Actual task effectiveness scores were calculated from average PVT inverse reaction time (pre-dose; 2 h post-dose; 6 h post-dose). Actigraphy-derived sleep intervals were used in SAFTE to determine modeled healthy control task effectiveness scores. **Results**: In participants with OSA (N = 31) on placebo or solriamfetol, actual and modeled healthy control task effectiveness did not differ at any time point. In participants with narcolepsy (N = 20) on placebo, actual task effectiveness at 2 h post-dose was lower than modeled healthy control task effectiveness (nominal P = 0.03), a difference not present with solriamfetol. There was no main effect of solriamfetol on actual or modeled healthy control task effectiveness across time points.

Conclusion: This study represents a novel application of the SAFTE biomathematical model to approximate healthy controls in sleep disorder research and provides valuable lessons that may optimize future research. Future studies should perform a priori power analyses for model-tested outcomes and use sleep measures that capture sleep fragmentation characteristic of sleep disorders for sleep input (e.g., total sleep time rather than time in bed).

Trial Registration: NCT02806895, EudraCT 2015-003930-28: A Randomized, Double-Blind, Placebo-Controlled, Crossover On-Road Driving Study Assessing the Effect of JZP-110 on Driving Performance in Subjects With Excessive Sleepiness Due to Obstructive Sleep Apnea. NCT02806908, EudraCT 2015-003931-36: A Randomized, Double-Blind, Placebo-Controlled, Crossover On-Road Driving Study Assessing the Effect of JZP-110 on Driving Performance in Subjects With Excessive Sleepiness Due to Narcolepsy.

L. P. Carter Alector, South San Francisco, CA, USA

L. P. Carter University of Arkansas for Medical Sciences, Little Rock, AR, USA

S. Overeem

Biomedical Diagnostics Group, Department of Electrical Engineering, Eindhoven University of Technology, Eindhoven, Netherlands

G. J. Lammers

Department of Neurology, Leiden University Medical Center, Leiden, Netherlands **Keywords:** Obstructive sleep apnea; Psychomotor vigilance task; Biomathematical modeling; SAFTE-FAST

Key Summary Points

Why carry out this study?

Excessive daytime sleepiness associated with obstructive sleep apnea or narcolepsy can impair attention and vigilance, which may lead to impaired driving performance and increase the risk for motor vehicle crashes.

Evaluating the effects of treatments for excessive daytime sleepiness, such as solriamfetol, on task performance is challenging because of differences in sleep history and sleep behavior between individuals.

These analyses explored the use of the Sleep, Activity, Fatigue, and Task Effectiveness (SAFTE) biomathematical model as a proxy for healthy controls using psychomotor vigilance task data from two placebo-controlled clinical trials of solriamfetol.

What was learned from the study?

This study represents a novel application of the SAFTE biomathematical model to approximate task effectiveness in healthy controls in sleep disorder research and provides valuable lessons that may optimize future research.

Future studies should perform a priori power analyses for model-tested outcomes and use sleep measures that capture sleep fragmentation characteristic of sleep disorders for sleep input (e.g., total sleep time rather than time in bed).

The use of SAFTE to predict performance in patients with narcolepsy may be limited due to the model's assumption of a normal relationship between sleep duration and changes in performance.

INTRODUCTION

Excessive daytime sleepiness (EDS) is a core symptom of narcolepsy [1] and has been reported to affect between 47% and 87% of patients with obstructive sleep apnea (OSA) at the time of their diagnosis [2, 3]. Further, EDS is reported to persist in 9-22% of patients with OSA despite treatment with continuous positive airway pressure [4, 5]. EDS associated with sleep disorders can impair vigilance and attention [6] and increases the risk for occupational injuries [7] and motor vehicle [8] crashes. Indeed, individuals with either narcolepsy or OSA are at increased risk of traffic collisions [9, 10], and a recent study of patients with OSA found that the presence of sleepiness rather than OSA severity (determined with the apnea-hypopnea index) determined this risk [11]. Wake-promoting agents have been demonstrated to reduce EDS in narcolepsy and OSA, and have been shown to improve real or simulated driving performance in controlled clinical studies [12, 13].

In clinical settings, attention and vigilance can be examined using a variety of measures, including the psychomotor vigilance task (PVT), a measure of sustained attention that correlates with subjective sleepiness [14]. Furthermore, sleep-related declines in PVT function correlate with poor performance on driving measures, such as lane drift [15]. Compared with healthy controls, patients with sleep disorders and sleep deprivation perform poorly on the PVT [16, 17]. However, some commonly used PVT metrics have been found to lack sensitivity, and differential vulnerability to poor performance on the PVT has been noted in a subset of healthy individuals [18, 19]. Additionally, measures commonly used to assess performance in clinical trials rarely overlap with those used in occupational settings.

When evaluating treatments for EDS secondary to sleep disorders, comparisons can be made between patients and healthy controls, and/or between patients receiving the treatment of interest and those receiving placebo, to determine the impact on performance in the target population. Ideally, comparisons would be made against a control group with the same sleep history; however, current approaches do not control for differences in sleep behavior (as either a function of disease history or drug effect) between individuals. Because differences in sleep behavior can affect outcomes of interest, such as measures of attention [20], the ability to isolate the specific effects of treatments on such outcomes can be confounded. Integrating biomathematical modeling, as used in occupational assessments of performance, may help compensate for shortcomings related to control of sleep behavior in clinical research.

The Sleep, Activity, Fatigue, and Task Effectiveness (SAFTE) biomathematical model has been used to predict the effects of fatigue on performance [21] in healthy individuals [22]. This model predicts cognitive performance based on several input parameters, including the accumulation of sleep over time, taking into account circadian processes and time of day. Pairing the SAFTE model with the Fatigue Avoidance Scheduling Tool (FAST) has provided a means (SAFTE-FAST) to estimate and mitigate exposure to fatigue risk throughout the day. The SAFTE model has widespread use among transportation (e.g., aviation, rail) agencies and the US Department of Defense, where it allows for the construction of schedules that avoid performance impairment attributable to fatigue [22]. The SAFTE model has previously been demonstrated to predict human factors-related freight rail accident risk based on predicted fatigue-induced impairments [23]. Analysis showed that the relative risk was increased by 42% when SAFTE-predicted effectiveness scores were < 77%, but was reduced by 30% when scores were > 90%. However, the SAFTE model has not been investigated in patients with sleep disorders. In the novel application described herein, the SAFTE model functioned as a proxy for healthy controls by predicting the treatment- and time-dependent effects that would have been expected had the participants been healthy controls with exactly the same sleep history as the participants from whom data were collected.

Solriamfetol (SunosiTM; Axsome Therapeutics, Inc., New York, NY), a dopamine and norepinephrine reuptake inhibitor, is approved in the USA [24] and European Union [25] to improve wakefulness in adults with EDS associated with narcolepsy (75-150 mg once daily) or OSA (37.5-150 mg once daily). Solriamfetol was shown to reduce EDS and improved quality of life in short-term (12 weeks) and longer term (up to 52 weeks) phase 3 clinical trials in participants with OSA or narcolepsy [26-32]. The effect of solriamfetol on real-world driving was recently examined in two randomized, crossover, placebo-controlled phase 2 trials in participants with narcolepsy or OSA [33, 34]. In studies, solriamfetol significantly both decreased (improved) standard deviation of lateral position (SDLP) at 2 h post-dose, meeting the primary efficacy endpoints. PVT and actigraphy data were also collected during these trials.

The aim of the analysis reported here was to explore the utility of the SAFTE model as a substitute for a healthy control group using predicted PVT performance based on actigraphy-derived sleep parameters compared with actual PVT performance from these two solriamfetol phase 2 clinical trials.

METHODS

Study Design

This analysis used data collected during two phase 2 studies examining the effects of solriamfetol on driving performance in participants (NCT02806895, with OSA EudraCT 2015-003930-28) or narcolepsy (NCT02806908, EudraCT 2015-003931-36) [33, 34]. The SAFTE modeling analyses were planned secondary efficacy endpoints for both trials. In these trials, participants were randomly assigned 1:1 to receive solriamfetol 150 mg/day for 3 days followed by 300 mg/day for 4 days, or placebo for all 7 days (Period 1), and then were crossed over to the other 7-day treatment (Period 2). As the studies were conducted before finalization of regulatory approval or dosing recommendations, the dose of 300 mg/day was based on prior phase 2 clinical trial results [35, 36] and aligned with the maximum dose used in the phase 3 clinical trials [26–28, 30].

These studies were conducted in accordance with the Declaration of Helsinki of 1964, and its later amendments. The study protocols were approved by the medical ethics committee of University Hospital Maastricht and Maastricht University (www.toetsingonline.nl; OSA: NL56214.068.16; narcolepsy: NL56215.068.16), and all participants provided written informed consent. The primary endpoint for both studies was the effect of solriamfetol on on-road driving performance, as assessed by SDLP at 2 h postdose (reported separately).

Participants

As reported previously [33, 34], participants ranged in age from 21 to 75 years, had either an OSA diagnosis per International Classification of Sleep Disorders-Third Edition (ICSD-3; American Academy of Sleep Medicine, Darien, IL, USA) or a narcolepsy diagnosis per ICSD-3 or Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition (American Psychiatric Association, Richmond, VA, USA), and an average total nightly sleep > 6 h. Additional inclusion criteria for participants with OSA included baseline Epworth Sleepiness Scale (ESS) score > 10, mean sleep latency < 30 min on the Maintenance of Wakefulness Test (MWT), and one of the following: use of a primary therapy for $OSA \ge 1$ night per week with < 2 days of variation week to week, lack of use of primary therapy after a history of > 1 month of attempted use with \geq 1 documented adjustment, or a history of surgical intervention. Participants who were unwilling to attempt to use > 1 primary OSA therapies were excluded.

Procedures

Participants were instructed to take a single capsule containing their treatment once daily, within 1 h of waking in the morning, on an empty stomach, and then to wait ≥ 30 min before having breakfast. At visits on days 7 and 14, participants completed the PVT along with an on-road driving test and other measures (reported separately) [33, 34]. On test days, the PVT was administered pre-dose and at

approximately 2 and 6 h post-dose. The PVT was administered over 10 min, with visual stimuli appearing at random variable intervals of 2–10 s; participants responded to a digital signal by pressing a key on a computer terminal. Participants also wore an actigraph from screening through day 14 (except during testing sessions, as data were extracted during that time) and kept accompanying sleep diaries.

Assessments and Outcomes

Participants completed the PVT by responding to a digital signal on a computer terminal via key presses, as described above. Calculated PVT measures included mean reaction time (MRT), lapses (reaction times > 500 ms or failures to respond), and errors (responses made without stimulus or false starts).

Sleep metrics, including time in bed (TIB), total sleep time (TST), and daily sleep intervals (DSIs), were measured at baseline and throughout the treatment period. Raw actigraphy data were scored using Actiware software (Philips Respironics, Bend, OR, USA) and manual scoring techniques [37]. Sleep diaries were used to assess the major sleep interval start and stop times and any naps not scored by the Philips algorithm but confirmed by activity pattern. For manual scoring, sleep diaries were compared with actigraphy data.

Task Effectiveness

Actual task effectiveness scores were calculated based on each participant's inverse reaction time (IRT [1/RT]) as described below. Sleep intervals derived from participants' actigraphy data—or from sleep diaries, if available, where actigraphy data were missing—were used to calculate modeled healthy control task effectiveness scores using the SAFTE [38, 39]. Participants evidenced good sleep hygiene, with minimal missing data. Task effectiveness was not modeled for participants missing data from the night before PVT testing. A SAFTE model effectiveness score of 100 indicates typical best performance during the day, based on a normal healthy population. SAFTE-FAST effectiveness predictions have been validated against IRT on the PVT [22].

Statistical Analyses

Target enrollment in the studies used to generate PVT data was based on the primary endpoint [33, 34]. SAFTE modeling was an experimental endpoint and as such was not considered for sample size calculations.

SAFTE-FAST modeled task effectiveness based on measured sleep served as a proxy for performance measures from healthy controls, referred to here as "modeled healthy control task effectiveness". Sleep intervals exported from the Actiware software-date and time of each recorded sleep interval's start and end-were used in SAFTE-FAST to determine a continuous estimate of modeled healthy control task effectiveness across the entire study period. As SAFTE-FAST generates a continuous estimate of task effectiveness, modeled healthy control task effectiveness scores from the time when PVTs were taken were identified using time stamp data from the PVT. Synchronized task effectiveness scores were exported from SAFTE-FAST for subsequent comparison against actual effectiveness.

Actual effectiveness scores were calculated as a percentage of a theoretical "best performance" from participants' average IRT per PVT trial (for all PVT test sessions, actual effectiveness = IRT/ 3.96×100). Average speed (IRT) on a PVT under optimal conditions—healthy sleepers taking the test at 11:00 in the morning, following an 8-h sleep opportunity—is approximately 3.96 [40], and is assumed to represent an actual effectiveness score of 100 (a healthy participant's typical best performance). Repeated-measures analysis of variance (ANOVA) was used to estimate the effect of treatment on actual effectiveness over time.

For additional analyses of data from model input parameters, repeated-measures ANOVA was performed to compare the effects of treatment on PVT measures (IRT, MRT, lapses, and errors). Fixed effects included treatment, time, and treatment \times time interaction. Treatment order; habitual TIB, TST, and DSIs; and previous day's TIB, TST, and DSIs were treated as covariates, while participant was modeled as a random effect. Post hoc tests (marginal means, pairwise comparisons) were performed to compare PVT measures between each trial. Mean habitual sleep parameters and sleep parameters obtained the day before PVT testing were compared across all conditions and with those obtained during the baseline period using Student's t tests, assuming unequal variance. The anticipated risk reduction associated with solriamfetol was computed as the difference between PVT performance metrics by treatment condition. All statistical analyses were performed with Stata 15.1 (StataCorp LLC, College Station, TX, USA) and Excel 2016 (Microsoft Corp., Redmond, WA, USA). P values were not controlled for multiplicity and therefore are nominal.

RESULTS

Demographics

For participants with OSA (safety population, n = 34; detailed demographics and disposition previously reported [33, 34]), 88% were male, mean (standard deviation [SD]) age was 51.6 (12.3) years, mean (SD) MWT sleep latency was 14.3 (7.3) min, and 85% used a primary OSA therapy.

For participants with narcolepsy (safety population, n = 24; detailed demographics and disposition previously reported [33, 34]), 54% were male, mean (SD) age was 40.4 (11.8) years, and mean (SD) MWT sleep latency was 4.0 (2.5) min (MWT latencies were based on historical data and were not collected as part of the study; n = 22).

Participants with missing sleep data (due to nonadherence or device errors) were excluded from analyses of PVT data and biomathematical modeling. Data were analyzed for 31 participants with OSA and 20 participants with narcolepsy. Use of SAFTE Biomathematical Modeling to Characterize Task Effectiveness Changes

Participants with OSA

In participants with OSA, analysis of actual task effectiveness showed no main effect of treatment or treatment × time interaction effects based on repeated-measures ANOVA (Table 1); however, there was an effect of previous day's sleep (P = 0.001). In a regression analysis, longer sleep the night before testing was not independently related to actual effectiveness ($R^2 = 0.01$, P = 0.98), suggesting this relationship was nonlinear. Actual task effectiveness was lower in participants treated with solriamfetol compared with placebo at pre-dose (P = 0.02) but not at 2 or 6 h post-dose (P > 0.05 for both).

In participants with OSA, repeated-measures ANOVA of modeled healthy control task effectiveness showed no main effect of treatment or treatment × time interaction effects; however, similarly to actual task effectiveness, there was an effect of previous day's sleep (P = 0.04). Longer sleep was independently related to higher modeled healthy control task effectiveness ($R^2 = 0.47$, P < 0.001).

There were no differences between modeled healthy control and actual task effectiveness scores for solriamfetol or for placebo in participants with OSA (P > 0.05 overall and for all time points) (Fig. 1).

Participants with narcolepsy

In participants with narcolepsy, repeated-measures ANOVA showed a treatment × time interaction effect (P < 0.001) on actual task effectiveness, with task effectiveness improving after administration of solriamfetol (from predose to 2 h post-dose to 6 h post-dose) but deteriorating after the administration of placebo (Table 1). Additionally, there was a main effect of previous day's sleep (P = 0.004), similar to that observed in participants with OSA. Longer sleep the night before testing was independently related to poorer actual effectiveness ($R^2 = 0.21$, P < 0.001)

	LS mean (SE)		LS mean	<i>P</i> value ^a	ANOVA P	values ^a	
-	Placebo	Solriamfetol	difference (95% CI)		Treatment	Treatment X Time	Previous day's sleep
OSA							
Actual task effectiveness ((n = 31)	(n = 31)			0.68	0.24	0.001
Pre-dose	93.64 (1.17)	89.19 (1.12)	- 4.45 (- 8.15, - 0.75)	0.02			
2 h post-dose	94.23 (1.17)	91.88 (1.12)	- 2.35 (- 6.06, 1.34)	0.21			
6 h post-dose	95.07 (1.17)	93.72 (1.12)	- 1.35 (- 5.05, 2.36)	0.47			
Modeled healthy control task effectiveness ((n = 31)	(n = 31)			0.10	0.92	0.04
Pre-dose 5	95.28 (0.19)	93.27 (0.18)	- 2.01 (- 2.62, - 1.41)	< 0.001			
2 h post-dose	96.13 (0.19)	94.11 (0.18)	- 2.02 (- 2.61, - 1.41)	< 0.001			
6 h post-dose	93.46 (0.19)	91.56 (0.18)	- 1.90 (- 2.51, - 1.31)	< 0.001			
Narcolepsy							
Actual task effectiveness ((n = 20)	(n = 20)			0.31	< 0.001	0.004
Pre-dose	92.23 (1.46)	86.06 (1.65)	- 6.16 (- 10.59, - 1.73)	0.007			
2 h post-dose {	85.66 (1.46)	96.07 (1.65)	$10.41 \ (5.97, \ 14.83)$	< 0.001			
6 h post-dose 8	86.29 (1.46)	93.89 (1.65)	7.60 (3.17, 12.03)	0.001			
Modeled healthy control task effectiveness ((n = 20)	(n = 20)			0.19	0.88	< 0.001
Pre-dose	94.39 (0.19)	93.49 (0.22)	-0.90(-1.48, -0.32)	0.003			
2 h post-dose	95.71 (0.19)	94.63 (0.22)	-1.08(-1.65, -0.49)	< 0.001			
6 h post-dose 5	93.41 (0.19)	92.35 (0.22)	-1.06(-1.64, -0.48)	< 0.001			



Fig. 1 Predicted and actual task effectiveness across PVT test sessions for participants with OSA under a Placebo conditions and b Solriamfetol conditions. Actual task effectiveness between drug conditions was compared with modeled healthy control task effectiveness using paired-

There was no main effect of treatment or treatment × time interaction effects on modeled healthy control task effectiveness based on repeated-measures ANOVA; however, in line with other analyses, there was an effect of previous day's sleep on modeled healthy control task effectiveness (P < 0.001). Longer sleep was independently related to higher modeled healthy control task effectiveness ($R^2 = 0.47$, P < 0.001).

In participants with narcolepsy, modeled healthy control task effectiveness was higher than actual task effectiveness for placebo (P = 0.009), primarily driven by PVT performance at 2 h post-dose (P = 0.03). Modeled healthy control and actual task effectiveness did not differ for solriamfetol (P > 0.05 overall and at all time points) (Fig. 2).

Change in SAFTE model input parameters and PVT measures

In participants with OSA or narcolepsy (Electronic Supplementary Material Figs. S1 and S2, respectively), there were no differences between placebo and solriamfetol treatment in either habitual sleep measures (TIB, TST, DSIs) or TIB,



samples *t* test, controlling for drug condition and test session. *P* values are nominal. ^aModeled healthy control vs. actual task effectiveness. *OSA* obstructive sleep apnea, *PVT* psychomotor vigilance task, *SD* standard deviation

TST, or DSIs measured the day before PVT testing.

There was no main effect of solriamfetol on any PVT measure in participants with OSA (Table 2; repeated-measures ANOVA, P > 0.05for all) or participants with narcolepsy (Table 3; repeated-measures ANOVA, P > 0.05 for all).

DISCUSSION

In this analysis, SAFTE modeling data (i.e., modeled healthy control task effectiveness) were used as a proxy for healthy participants in a clinical trial. This represents a novel application of SAFTE, which previously has been used to model the effects of sleep deprivation in healthy populations [21, 22]. In participants with OSA, actual task effectiveness did not differ from modeled healthy control task effectiveness with placebo or with solriamfetol, despite participants' EDS (evidenced by a MWT mean sleep latency of 14 min at baseline). In light of this finding (i.e., an apparent lack of a deficit in performance in participants with OSA), it is unsurprising that actual task effectiveness did not differ between placebo and solriamfetol in participants with OSA. In participants with



Fig. 2 Predicted and actual task effectiveness across PVT test sessions for participants with narcolepsy under (a) Placebo conditions and (b) Solriamfetol conditions. Actual task effectiveness between drug conditions was compared with modeled healthy control task effectiveness using

narcolepsy, actual task effectiveness with placebo was lower than modeled healthy control task effectiveness (suggesting a deficit in performance in participants with narcolepsy), whereas actual task effectiveness with solriamfetol did not differ from modeled healthy control effectiveness; actual task effectiveness improved with solriamfetol relative to placebo. Thus, in contrast to the differences in SDLP between solriamfetol and placebo (reported elsewhere), the present findings suggest that, in these participants with OSA, task effectiveness while on placebo or solriamfetol was similar to that of "healthy controls", despite the participants' sleep disorder. In participants with narcolepsy, however, task effectiveness while on placebo was lower than that of "healthy controls" and improved on solriamfetol to a level similar to that of "healthy controls."

Analyses of the primary outcomes of these studies demonstrated that solriamfetol improved real-world on-road driving performance, as measured by SDLP, at 2 h post-dose in participants with narcolepsy as well as in participants with OSA. Therefore, the SAFTE model findings can be considered to be generally consistent with the primary study findings for participants with narcolepsy. In contrast, the



paired-samples t test, controlling for drug condition and test session. P values are nominal. ^aModeled healthy control vs. actual task effectiveness. PVT, psychomotor vigilance task, SD, standard deviation

results for participants with OSA suggest that, despite their degree of EDS, these individuals had a relative lack of impairment in PVT performance and task effectiveness scores while on placebo. Although the small sample sizes preclude any definitive conclusions, it is possible that the lack of detectable improvement may be due to a "ceiling effect." This finding may indicate that PVT performance is not sensitive enough to detect impairment attributable to EDS in OSA in a study of this size.

There was no effect of solriamfetol treatment on any of the sleep measures recorded in participants with OSA or participants with narcolepsy. This finding aligns with data from phase 3 trials of solriamfetol in participants with narcolepsy and OSA in which sleep measured by polysomnography was unchanged with solriamfetol compared with placebo [25, 26]. However, as sleep is a fundamental input for the SAFTE model, the lack of an effect of solriamfetol on sleep measures is a key consideration in interpreting the present findings.

The present analysis found no effect of solriamfetol on PVT metrics in participants with OSA; however, participants with narcolepsy were faster (IRT) and had fewer lapses after treatment with solriamfetol compared with

Table 2 Results of psyc	chomotor vigilance	e task in participants	with obstructive sleep apnea				
PVT parameters	LS mean (SE)		LS mean difference (95%	P value ^a	A NOVA P	values ^a	
	$\frac{\text{Placebo}}{(n=31)}$	Solriamfetol (n = 31)	- CI)		Treatment	Treatment × Time	Previous day's sleep
Inverse reaction time, 1/ms					0.68	0.24	0.001
Pre-dose	3.71 (0.05)	3.53(0.04)	- 0.18 (- 0.32, - 0.03)	0.02			
2 h post-dose	3.73 (0.05)	3.64 (0.04)	- 0.09 (- 0.24, 0.05)	0.21			
6 h post-dose	3.76 (0.05)	3.71 (0.04)	- 0.05 (- 0.20, 0.09)	0.47			
Mean reaction time, ms					0.31	0.25	0.02
Pre-dose	296 (6.22)	303 (5.94)	6.5 (- 13.2, 26.2)	0.52			
2 h post-dose	296 (6.22)	292 (5.94)	- 3.96 (- 23.7, 15.7)	0.69			
6 h post-dose	302 (6.22)	292 (5.94)	- 9.84 (- 29.5, 9.9)	0.32			
Lapses, <i>n</i> /trial					0.34	0.91	0.49
Pre-dose	5.95 (0.94)	3.82~(0.90)	- 2.13 (- 5.11, 0.85)	0.16			
2 h post-dose	5.66 (0.94)	3.02~(0.90)	- 2.64 (- 5.62, 0.34)	0.08			
6 h post-dose	6.04 (0.94)	3.98(0.90)	- 2.06 (- 5.04, 0.92)	0.17			
Errors, $n/$ trial					0.78	0.77	1.00
Pre-dose	1.10(0.61)	1.24(0.58)	0.14(-1.79, 2.07)	0.89			
2 h post-dose	2.39 (0.61)	$1.84\ (0.58)$	- 0.55 (- 2.48, 1.38)	0.57			
6 h post-dose	1.91 (0.61)	$1.84\ (0.58)$	- 0.07 (- 2.00, 1.86)	0.95			
<i>ANOVA</i> analysis of vari ^a <i>P</i> values are nominal	iance, <i>CI</i> confiden	ice interval, LS least s	squares, OSA obstructive sleep	apnea, PVT	⁷ psychomotc	r vigilance task, <i>SE</i> sta	ındard error

I able 5 Kesults of psyc	homotor vigilance	e task in participants v	with narcolepsy				
PVT parameters	LS mean (SE)		LS mean difference (95%	<i>P</i> value ^a	A NOVA P	values ^a	
	Placebo (n = 20)	Solriamfetol (n = 20)	- CI)		Treatment	Treatment × Time	Previous day's sleep
Inverse reaction time, <u>1/ms</u>					0.31	< 0.001	0.004
Pre-dose	3.65 (0.06)	3.41 (0.07)	-0.24(-0.42, -0.07)	0.007			
2 h post-dose	3.39~(0.06)	3.80 (0.07)	$0.41 \ (0.24, \ 0.59)$	< 0.001			
6 h post-dose	3.42 (0.06)	3.72 (0.07)	$0.30\ (0.13,\ 0.48)$	0.001			
Mean reaction time, ms					0.52	0.06	0.91
Pre-dose	388 (131.2)	582 (148.4)	194 (- 204.2, 591.4)	0.34			
2 h post-dose	699 (131.2)	317 (148.4)	- 382 (- 779.9, 15.7)	0.06			
6 h post-dose	657 (131.2)	319 (148.4)	- 338 (- 735.8, 59.8)	0.10			
Lapses, n/trial					0.59	0.03	0.65
Pre-dose	4.36(1.42)	6.63(1.61)	2.27 (- 2.04, 6.58)	0.30			
2 h post-dose	7.91 (1.42)	3.35 (1.61)	- 4.56 (- 8.87, - 0.25)	0.04			
6 h post-dose	8.96 (1.42)	4.52 (1.61)	- 4.44 (- 8.75, - 0.13)	0.04			
Errors, <i>n</i> /trial					0.50	0.92	1.00
Pre-dose	1.53(0.37)	1.11(0.42)	- 0.42 (- 1.55, 0.73)	0.47			
2 h post-dose	$1.73 \ (0.37)$	1.39(0.42)	-0.34(-1.47, 0.80)	0.57			
6 h post-dose	1.98(0.37)	1.34 (0.32)	-0.64(-1.78, 0.49)	0.26			
ANOVA analysis of vari	ance, CI confiden	ce interval, LS least se	quares, OSA obstructive sleep a	apnea, <i>PVT</i>	psychomotor	: vigilance task, SE stan	ıdard error

placebo. These results are consistent with the SAFTE findings. However, previous studies established the efficacy of solriamfetol in reducing EDS in patients with OSA or narcolepsy [26, 28, 30], and the primary results of the present studies demonstrated that solriamfetol improved real-world on-road driving performance, as measured by SDLP, at 2 h postdose in both groups of participants. As this study was powered for SDLP rather than PVT or SAFTE, it may be that larger studies are needed to observe further differences in participants

with OSA. Several factors should be kept in mind when considering the PVT findings in particular. Most important, these studies had small sample sizes and were not powered to detect a difference in PVT scores with solriamfetol treatment. In patients with OSA, PVT scores were previously found not to differ by apnea-hypopnea index severity [41], and have been shown to be associated with subjective (ESS) but not objective (MWT) sleepiness in patients with OSA [14]. Further, PVT performance has been found to be worse in patients with narcolepsy than in patients with insufficient sleep syndrome [16].

The real-world driving performance, as measured by SDLP (an established correlate of drug- and alcohol-induced motor vehicle crash risk [42]), was improved at 2 h post-dose in both studies [33, 34]. SAFTE-modeled healthy control task effectiveness was not examined in the context of real-world driving performance in healthy controls in this study or in previous studies. The SAFTE model was not designed as a performance predictor for SDLP. If in future studies investigators intend to use a biomathematical model in lieu of a healthy control population, power analyses should be performed to reflect PVT performance or similar model-tested performance outcomes a priori. The SAFTE model has not been optimized for the prediction of performance or risk in people with sleep disorders.

While the goal for the current analysis was to use SAFTE-FAST as a proxy for healthy controls, some lessons can be extrapolated to the application of biomathematical modeling to sleep disorders. For this analysis, the model used patient TIB and assumed excellent sleep quality, as the goal of the model was to serve as a proxy for healthy controls. Patients with sleep disorders often have sleep fragmentation (e.g., because of apneas, as in patients with OSA), which may not be apparent in actigraphy data and is not considered when computing TIB. SAFTE-FAST can use TST instead of TIB for a sleep input, and users can modify the sleep quality setting to reflect disrupted sleep. Using TST or adjusting the sleep quality of sleep events could help replicate the objective fragmentation of sleep seen in sleep disorders. One area of limitation is that the model assumes a normal relationship between sleep duration and changes in performance, but this relationship may be weaker in patients with narcolepsy [43]. The weakening of this relationship may explain why longer prior day's sleep was not independently correlated with improved actual task effectiveness in either group. The biological underpinnings that could explain the breakdown between sleep and performance in narcolepsy need to be better understood before a biomathematical model can reasonably be developed.

This novel application of the SAFTE model in a clinical trial setting highlights the need for additional metrics and thereby informs future studies. The crossover design used here strengthens the validity of its comparisons by controlling for interindividual differences between treatment conditions. Indeed. although PVT and SAFTE-FAST provide useful safety benchmarks, neither PVT nor SAFTE-FAST task effectiveness scores are used in isolation as determinants of ability to drive or perform other tasks. Aviation and other industries want to limit the amount of time spent below 77% effectiveness, which is equivalent to a 0.05% blood alcohol concentration [44]. In the present analysis of participants with OSA or narcolepsy, both populations had actual task effectiveness levels above this cutoff based on the model as implemented; however, these findings should not be over-interpreted as an equivocal statement of safety, especially given the extensive published literature demonstrating the increased risk of motor vehicle crashes in these populations [9, 10, 45–48].

There are several limitations to this study. First, the exclusion of participants who did not habitually sleep ≥ 6 h per night may limit the interpretation of the effect of solriamfetol on participants whose sleep patterns are severely disrupted by OSA or narcolepsy. Therefore, it is possible that a more broadly defined population of patients with OSA or narcolepsy with varied sleep tendencies would have yielded different results. Second, there was no washout period between treatments, thus there may have been carryover effects of solriamfetol. However, testing did not occur until day 7 of each treatment period. Solriamfetol is rapidly absorbed (median T_{max} : appprox. 2 h), with a mean elimination half-life of approximately 6 h [49], which suggests that plasma levels would be negligible by day 7 after the final dose of solriamfetol. Thus, the study design effectively created a 7-day "washout" for participants who crossed over from solriamfetol to placebo. Further, a previous phase 2 trial in participants with narcolepsy did not suggest carryover effects when demonstrating the efficacy of solriamfetol using a crossover design with no washout between treatment periods [35]. Third, there are no published task effectiveness standards for patients with OSA or narcolepsy, thus there is no basis in the literature for comparison of the present results. Fourth, actual healthy control data were not collected in this study to compare against SAFTE prediction of modeled healthy control task effectiveness, or to compare sleep quality against SAFTE assumptions about healthy control sleep history or PVT performance. Fifth, PVT was the only attention measure used in the analysis. Since PVT performance does not always predict impaired attention associated with sleep deprivation, even among healthy participants [18], inclusion of additional attention metrics may have provided additional context with respect to drug effects and performance. Finally, all participants received 300 mg/day of solriamfetol in the final 4 days of the active treatment period; therefore, any impact of solriamfetol on task effectiveness would reflect effects of the 300 mg/day dose. The maximum recommended dose of solriamfetol in the US and EU is 150 mg/day.

CONCLUSIONS

This study represents a novel application of the SAFTE-FAST biomathematical model as a substitution for healthy controls in a research investigation of patients with sleep disorders. The results provide valuable lessons that may help to optimize future studies that incorporate this model to ensure the applicability of their results. For example, future studies should examine the use of sleep measures (other than TIB) that reflect the objective fragmentation of sleep seen in sleep disorders. This analysis also provides additional context for how PVT performance during a clinical trial may differ from performance on real-world performance measures, such as lane drift.

ACKNOWLEDGEMENTS

Funding. This study was supported by Jazz Pharmaceuticals (Dublin, Ireland). At the time that the study was conducted, Jazz Pharmaceuticals had worldwide development, manufacand commercialization rights to turing. solriamfetol, excluding certain jurisdictions in Asia. Jazz Pharmaceuticals completed the divestiture of Sunosi® (solriamfetol) to Axsome Therapeutics, Inc. in the US on May 9, 2022 and ex-US on November 14, 2022. SK Biopharmaceuticals (Seongnam, South Korea), the discoverer of the compound (also known as SKL-N05), maintains rights in 12 Asian markets, including Korea, China, and Japan. The journal's Rapid Service Fee was funded by Axsome Therapeutics.

Authorship. All named authors meet the International Committee of Medical Journal Editors (ICMJE) criteria for authorship for this article, take responsibility for the integrity of the work as a whole, and have given their approval for this version to be published.

Medical Writing, Editorial, and Other Assistance. Under the direction of the authors, Sherri Jones, PharmD and Christopher Jaworski of Peloton Advantage, LLC, an OPEN Health company, provided medical writing and editorial support for this article, which was funded by Jazz Pharmaceuticals and Axsome Therapeutics. Kefei Zhou (a former employee of Jazz Pharmaceuticals, Palo Alto, CA, USA) performed statistical analyses for these studies.

Author Contributions. Study design: S Hursh, A Vermeeren, IG Ramaekers, D Chen, L Carter, GJ Lammers. Study investigators: A Vermeeren, C Donjacour, JG Ramaekers, S Overeem, GJ Lammers. Enrollment of patients: J Asin, N de Vries, P Vonk, C Donjacour, H Janssen, S Overeem, JG Lammers. Collection and assembly of data: L Schwartz, N de Vries, P Vonk, F Vinckenbosch, JG Ramaekers, G Wang, S Overeem, JG Lammers. Data analysis: JK Devine, L Schwartz, S Hursh, JG Ramaekers, G Wang. Data interpretation: all authors. Manuscript preparation: JK Devine, L Schwartz, C Donjacour, G Wang. Manuscript review and revisions: all authors. Final approval of manuscript: all authors.

Disclosures. Jaime K Devine and Lindsay Schwartz are affiliated with the Institutes for Behavior Resources, which provides sales of SAFTE-FAST software, but do not benefit financially or non-financially from sales of SAFTE-FAST software. Steven Hursh is affiliated with the Institutes for Behavior Resources, which provides sales of SAFTE-FAST software. Steven Hursh is the inventor of the SAFTE-FAST biomathematical model, and a fraction of his compensation is based on sales of the software. Jerryll Asin has nothing to disclose. Nicolaas de Vries has been an advisory board participant for NightBalance, a study investigator for Inspire, and a consultant for the AE Mann Foundation, Olympus, and Philips. Patty E Vonk has nothing to disclose. Annemiek Vermeeren is an employee of Maastricht University. Maastricht University received financial support to conduct the present study. Claire EHM Donjacour has received a research grant from UCB Pharma, and served as a paid speaker for UCB Pharma and Eisai. Frederick Vinckenbosch is an employee of Maastricht University. Maastricht University received financial support to conduct the present study. Johannes G Ramaekers has received grants from pharmaceutical industries as well as national (NWO, ZonMw) and international (EU Commission) funding bodies that are unrelated to this work. Hennie Janssen has nothing to disclose. Grace Wang, Dan Chen, and Lawrence P Carter are former employees of Jazz Pharmaceuticals who, in the course of their employment, received stock options exercisable for, and other stock awards of, ordinary shares of Jazz Pharmaceuticals plc. Sebastiaan Overeem has received an unrestricted grant from UCB Pharma for research unrelated to this work and has served on advisory boards for UCB Pharma. Jazz Pharmaceuticals, Bioprojet, and Takeda, all paid to the institution. Gert Jan Lammers has received consultancy fees and/or honoraria and has been a speakers' bureau member and/or an advisory board participant for UCB Pharma, Bioprojet. Theranexus. and Iazz Pharmaceuticals.

Compliance With Ethics Guidelines. These studies were conducted in accordance with the Declaration of Helsinki of 1964, and its later amendments. The study protocols were approved by the medical ethics committee of University Hospital Maastricht and Maastricht University (www.toetsingonline.nl; OSA, NL56214.068.16; narcolepsy, NL56215.068.16), and all participants provided written informed consent.

Data Availability. All relevant data are provided within the manuscript and supporting files.

Open Access. This article is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License, which permits any non-commercial use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/bync/4.0/.

REFERENCES

- 1. Gandhi KD, Mansukhani MP, Silber MH, Kolla BP. Excessive daytime sleepiness: a clinical review. Mayo Clin Proc. 2021;96(5):1288–301.
- Seneviratne U, Puvanendran K. Excessive daytime sleepiness in obstructive sleep apnea: prevalence, severity, and predictors. Sleep Med. 2004;5(4): 339–43.
- 3. Bjorvatn B, Lehmann S, Gulati S, Aurlien H, Pallesen S, Saxvig IW. Prevalence of excessive sleepiness is higher whereas insomnia is lower with greater severity of obstructive sleep apnea. Sleep Breath. 2015;19(4):1387–93.
- 4. Gasa M, Tamisier R, Launois SH, Sapene M, Martin F, Stach B, et al. Residual sleepiness in sleep apnea patients treated by continuous positive airway pressure. J Sleep Res. 2013;22(4):389–97.
- Pepin JL, Viot-Blanc V, Escourrou P, Racineux JL, Sapene M, Levy P, et al. Prevalence of residual excessive sleepiness in CPAP-treated sleep apnoea patients: the French multicentre study. Eur Respir J. 2009;33(5):1062–7.
- 6. Van Schie MK, Thijs RD, Fronczek R, Middelkoop HA, Lammers GJ, Van Dijk JG. Sustained attention to response task (SART) shows impaired vigilance in a spectrum of disorders of excessive daytime sleepiness. J Sleep Res. 2012;21(4):390–5.
- Gharibi V, Mokarami H, Cousins R, Jahangiri M, Eskandari D. Excessive daytime sleepiness and safety performance: comparing proactive and reactive approaches. Int J Occup Environ Med. 2020;11(2):95–107.
- Ward KL, Hillman DR, James A, Bremner AP, Simpson L, Cooper MN, et al. Excessive daytime sleepiness increases the risk of motor vehicle crash in obstructive sleep apnea. J Clin Sleep Med. 2013;9(10):1013–21.
- 9. Lloberes P, Levy G, Descals C, Sampol G, Roca A, Sagales T, et al. Self-reported sleepiness while driving as a risk factor for traffic accidents in patients

with obstructive sleep apnoea syndrome and in non-apnoeic snorers. Respir Med. 2000;94(10): 971–6.

- 10. Philip P, Sagaspe P, Lagarde E, Leger D, Ohayon MM, Bioulac B, et al. Sleep disorders and accidental risk in a large group of regular registered highway drivers. Sleep Med. 2010;11(10):973–9.
- 11. Philip P, Bailly S, Benmerad M, Micoulaud-Franchi JA, Grillet Y, Sapène M, et al. Self-reported sleepiness and not the apnoea hypopnoea index is the best predictor of sleepiness-related accidents in obstructive sleep apnoea. Sci Rep. 2020;10(1): 16267.
- 12. Philip P, Chaufton C, Taillard J, Capelli A, Coste O, Leger D, et al. Modafinil improves real driving performance in patients with hypersomnia: a randomized double-blind placebo-controlled crossover clinical trial. Sleep. 2014;37(3):483–7.
- 13. Kay GG, Feldman N. Effects of armodafinil on simulated driving and self-report measures in obstructive sleep apnea patients prior to treatment with continuous positive airway pressure. J Clin Sleep Med. 2013;9(5):445–54.
- 14. Li Y, Vgontzas A, Kritikou I, Fernandez-Mendoza J, Basta M, Pejovic S, et al. Psychomotor vigilance test and its association with daytime sleepiness and inflammation in sleep apnea: clinical implications. J Clin Sleep Med. 2017;13(9):1049–56.
- 15. Baulk SD, Biggs SN, Reid KJ, van den Heuvel CJ, Dawson D. Chasing the silver bullet: measuring driver fatigue using simple and complex tasks. Accid Anal Prev. 2008;40(1):396–402.
- 16. Thomann J, Baumann CR, Landolt HP, Werth E. Psychomotor vigilance task demonstrates impaired vigilance in disorders with excessive daytime sleepiness. J Clin Sleep Med. 2014;10(9):1019–24.
- 17. Jung CM, Ronda JM, Czeisler CA, Wright KP Jr. Comparison of sustained attention assessed by auditory and visual psychomotor vigilance tasks prior to and during sleep deprivation. J Sleep Res. 2011;20(2):348–55.
- 18. Chua EC, Sullivan JP, Duffy JF, Klerman EB, Lockley SW, Kristal BS, et al. Classifying attentional vulnerability to total sleep deprivation using baseline features of psychomotor vigilance test performance. Sci Rep. 2019;9(1):12102.
- 19. Basner M, Dinges DF. Maximizing sensitivity of the psychomotor vigilance test (PVT) to sleep loss. Sleep. 2011;34(5):581–91.
- 20. Santhi N, Horowitz TS, Duffy JF, Czeisler CA. Acute sleep deprivation and circadian misalignment

associated with transition onto the first night of work impairs visual selective attention. PLoS ONE. 2007;2(11): e1233.

- 21. Hursh SR, Redmond DP, Johnson ML, Thorne DR, Belenky G, Balkin TJ, et al. Fatigue models for applied research in warfighting. Aviat Space Environ Med. 2004;75(3 suppl):A44–53.
- 22. Hursh S, Balkin T, Miller J, Eddy D. The Fatigue Avoidance Scheduling Tool: modeling to minimize the effects of fatigue on cognitive performance. SAE Tech Paper. 2004;2004-01-2151.
- 23. Hursh S, Fanzone JF, Raslear TG. Analysis of the relationship between operator effectiveness measures and economic impacts of rail accidents Washington DC: Federal Railroad Administration. 2011. https://railroads.dot.gov/elibrary/analysis-relationship-between-operator-effectiveness-measures-and-economic-impacts-rail. Accessed 1 Dec 2022.
- 24. Axsome Therapeutics, Inc. SunosiTM (solriamfetol) tablets prescribing information. New York: Axsome Therapeutics, Inc. 2022. https://www.sunosi.com/pi/sunosi.en.USPI.pdf. Accessed 30 Nov 2022.
- 25. Jazz Pharmaceuticals Ireland Ltd. Sunosi[™] (solriamfetol) tablets Summary of Product Characteristics. Dublin: Jazz Pharmaceuticals Ireland Ltd. 2020. https://www.ema.europa.eu/en/documents/ product-information/sunosi-epar-productinformation_en.pdf. Accessed 30 Nov 2022.
- 26. Schweitzer PK, Rosenberg R, Zammit GK, Gotfried M, Chen D, Carter LP, et al. Solriamfetol for excessive sleepiness in obstructive sleep apnea (TONES 3): a randomized controlled trial. Am J Respir Crit Care Med. 2019;199(11):1421–31.
- 27. Strollo PJ Jr, Hedner J, Collop N, Lorch DG Jr, Chen D, Carter LP, et al. Solriamfetol for the treatment of excessive sleepiness in OSA: a placebo-controlled randomized withdrawal study. Chest. 2019;155(2): 364–74.
- 28. Thorpy MJ, Shapiro C, Mayer G, Corser BC, Emsellem H, Plazzi G, et al. A randomized study of solriamfetol for excessive sleepiness in narcolepsy. Ann Neurol. 2019;85(3):359–70.
- 29. Emsellem HA, Thorpy MJ, Lammers GJ, Shapiro CM, Mayer G, Plazzi G, et al. Measures of functional outcomes, work productivity, and quality of life from a randomized, phase 3 study of solriamfetol in participants with narcolepsy. Sleep Med. 2020;67: 128–36.
- Malhotra A, Shapiro C, Pepin JL, Hedner J, Ahmed M, Foldvary-Schaefer N, et al. Long-term study of the safety and maintenance of efficacy of

solriamfetol (JZP-110) in the treatment of excessive sleepiness in participants with narcolepsy or obstructive sleep apnea. Sleep. 2020;43(2):zsz220.

- 31. Weaver TE, Drake CL, Benes H, Stern T, Maynard J, Thein SG, et al. Effects of solriamfetol on quality of life measures from a 12-week phase 3 randomized trial. Ann Am Thorac Soc. 2020;17(8):998–1007.
- 32. Weaver TE, Pepin JL, Schwab R, Shapiro C, Hedner J, Ahmed M, et al. Long-term effects of solriamfetol on quality of life and work productivity in participants with excessive daytime sleepiness associated with narcolepsy or obstructive sleep apnea. J Clin Sleep Med. 2021;17(10):1995–2007.
- 33. Vinckenbosch F, Lammers G, Overeem S, Chen D, Wang G, Carter L, et al. Effects of solriamfetol on on-the-road driving in participants with narcolepsy: a randomised crossover trial. Hum Psychopharmacol Clin Exp. 2022. https://doi.org/10. 1002/hup.2858. Accessed 1 Dec 2022.
- 34. Vinckenbosch F, Asin J, de Vries N, Vonk PE, Donjacour C, Lammers GJ, et al. Effects of solriamfetol on on-the-road driving performance in participants with excessive daytime sleepiness associated with obstructive sleep apnoea. Hum Psychopharmacol. 2022;37(6):e2845.
- 35. Bogan RK, Feldman N, Emsellem HA, Rosenberg R, Lu Y, Bream G, et al. Effect of oral JZP-110 (ADX-N05) treatment on wakefulness and sleepiness in adults with narcolepsy. Sleep Med. 2015;16(9): 1102–8.
- 36. Ruoff C, Swick TJ, Doekel R, Emsellem HA, Feldman NT, Rosenberg R, et al. Effect of oral JZP-110 (ADX-N05) on wakefulness and sleepiness in adults with narcolepsy: a phase 2b study. Sleep. 2016;39(7): 1379–87.
- 37. Ancoli-Israel S, Martin JL, Blackwell T, Buenaver L, Liu L, Meltzer LJ, et al. The SBSM guide to actigraphy monitoring: clinical and research applications. Behav Sleep Med. 2015;13(suppl 1):S4–38.
- 38. Rawlings Hursh S, Elsmore T, Eddy D, Inventors. Interface for a system and method for evaluating task effectiveness based on sleep patterns. US patent 7,118,530. October 10, 2006: Science Applications International Corp., assignee. 2006. https://patents. google.com/patent/US7118530B2/en. Accessed 1 Dec 2022.
- Rawlings Hursh S. Evaluating task effectiveness based on sleep pattern. US patent 7,2007,938. April 24, 2007: Science Applications International Corporation, assignee. 2007. https://patents.google. com/patent/US7207938B2/en. Accessed 1 Dec 2022.

- 40. Jongen S, Perrier J, Vuurman EF, Ramaekers JG, Vermeeren A. Sensitivity and validity of psychometric tests for assessing driving impairment: effects of sleep deprivation. PLoS ONE. 2015;10(2): e0117045.
- 41. Batool-Anwar S, Kales SN, Patel SR, Varvarigou V, DeYoung PN, Malhotra A. Obstructive sleep apnea and psychomotor vigilance task performance. Nat Sci Sleep. 2014;6:65–71.
- 42. Owens K, Ramaekers JG. Drugs, driving, and models to measure driving impairment. In: Verster JC, Pandi-Perumal SR, Ramaekers JG, de Gier JJ, editors. Drugs, driving and traffic safety. Basel: Birkhauser Verlag; 2009. p. 43–58.
- 43. Broughton WA, Broughton RJ. Psychosocial impact of narcolepsy. Sleep. 1994;17(8 suppl):S45–9.
- 44. Federal Aviation Administration (FAA), US Department of Transportation (DOT). Flightcrew member duty and rest requirements: final rule. Fed Regist. 2012;77(2):329–403.
- 45. Liu SY, Perez MA, Lau N. The impact of sleep disorders on driving safety–findings from the Second

Strategic Highway Research Program naturalistic driving study. Sleep. 2018;41(4):1–11.

- Pizza F, Jaussent I, Lopez R, Pesenti C, Plazzi G, Drouot X, et al. Car crashes and central disorders of hypersomnolence: a French study. PLoS ONE. 2015;10(6):e0129386.
- 47. Tzeng NS, Hsing SC, Chung CH, Chang HA, Kao YC, Mao WC, et al. The risk of hospitalization for motor vehicle accident injury in narcolepsy and the benefits of stimulant use: a nationwide cohort study in Taiwan. J Clin Sleep Med. 2019;15(6): 881–9.
- 48. Tregear S, Reston J, Schoelles K, Phillips B. Obstructive sleep apnea and risk of motor vehicle crash: systematic review and meta-analysis. J Clin Sleep Med. 2009;5(6):573–81.
- 49. Zomorodi K, Kankam M, Lu Y. A phase I, randomized, crossover, open-label study of the pharmacokinetics of solriamfetol (JZP-110) in healthy adult subjects with and without food. Clin Ther. 2019;41(2):196–204.